responses letter

Dear Editor:

Thank you for sending me the comments on our submitted manuscript (Manuscript NO.: 90865, Editorial). According to your suggestion, we have considered all of the suggestions proposed by the reviewers and editor, and revised the manuscript thoroughly. The following are our point-to-point responses (referee’s comments are shown in italic):

Responses to Reviewer 1

Comment 1: I came across minor spelling/grammer mistakes. Page 5 line should read "elevated crp was observed in several" instead of "elevated crp was observed in sevral". Page 8 line 1 "steroid begins tapering when...." would need correction in it's formation. Thank you.

Answer 1: Thank you very much for your constructive comment and suggestion. We have corrected "elevated crp was observed in sevral...." into "Elevated C-reactive protein (CRP) was observed in several patients...." in Page 5 line 5. We have corrected "steroid begins tapering when...." into "The use of steroids, most commonly prednisone, begins to taper when clinical symptoms reduce to grade 1....". Revised portions are highlighted with yellow color in the revised manuscript. We appreciate your kind support to our manuscript. Thank you!

Responses to Editor

Comment 1: The main text of the manuscript must include two parts: "Introduction" and "Conclusion".

Answer 1: Thank you very much for your constructive comment and suggestion. We have made corrections, "Introduction" and "Conclusion" are included in the main text of the revised manuscript.

We provide English Language Certificate issued by AJE and have uploaded. We appreciate your kind support to our manuscript. We have made a point-by-point
response to the comments as above and highlight all the amends on manuscript. Thank you very much.